Author's response to reviews

Title: Insulin Use and Persistence in Patients with Type 2 Diabetes Adding Mealtime Insulin to a Basal Regimen: A Retrospective Database Analysis

Authors:

Machaon M Bonafede (machaon.bonafede@thomsonreuters.com)
Anupama Kalsekar (kalsekar_anupama@lilly.com)
Manjiri Pawaskar (pawaskar_manjiri@lilly.com)
Kimberly M Ruiz (kimberly.ruiz@thomsonreuters.com)
Amelito M Torres (lito.torres@thomsonreuters.com)
Karen R Kelly (kelly_karen_rae@lilly.com)
Suellen M Curkendall (suellen.curkendall@thomsonreuters.com)

Version: 3 Date: 6 January 2011

Author's response to reviews: see over
January 6, 2010

To Whom It May Concern:

We are pleased to submit the accompanying revised manuscript, titled, “Insulin Use and Persistence in Patients with Type 2 Diabetes Adding Mealtime Insulin to a Basal Regimen: A Retrospective Database Analysis” to BioMed Central. We have addressed the list and major and minor formatting revisions. Please do not hesitate to let me know if there are additional formatting issues that we can address or we misinterpreted any of the instructions.

Sincerely,

Mac Bonafede (corresponding author)

Machaon Bonafede, Ph.D. MPH
Research Leader

Thomson Reuters
37 Lowell Street
Andover, MA 01810
Mobile:585.766.8622

Thomson Healthcare is now the Healthcare business of Thomson Reuters.